Accelerator Status Report

Run Schedule: Jan. 3, 2011 through May 13, 2011
March 2, 2011
Injector    Energy 61.76 MeV
· The injector has been alright. Sometimes beam orbits do not reproduce after the gun HV off. The injector steering script should be used to restore the orbits.

· (2/25) The laser spot was moved to boost the QE. 
· (2/22) The polarization flipper was changed from ‘flip’ left to ‘flip’ right. The injector was optimized.
· (2/21) Control modules of cavities 0L04-5 & 8 were replaced. The 45MeV spectrometer was used to phase the cavities and the spectrometer was left on (~6 hrs.).

· (2/17) Both Capture 2.5W Amplifiers were replaced. After the replacement, no adjustment was needed to transport beam (~1.5 hrs.). 
Accelerator   Linac Energy 549.00 MeV
· The optics was stable but beam availability is about 50% due to the power interruptions and a CHL trip.

· (3/1) Did some beam studies trying to improve beam quality for Hall C. 

· (3/1) ESR tripped during a cable checkup (1.85 hrs.). 

· (2/26) The two way 1-pass split was tried but not successful. Box supplies do not regulate well at low current.

· (2/25) Around 0420 there was power interruption caused by Virginia Power maintenance work outside the Lab. After most systems were recovered VBV2L22B would not open. The valve would not open because the power cord was left disconnected during the vacuum work in the region. The valve was jumpered open and the jumper needs to be removed since the power cable was connected. We lost more than 21 hours. 
· The main machine has been stable.

· (2/22) ITV6E02 failed and need to be repaired.
· (2/22) CHL tripped around 1100 due to a user error during CHL maintenance and tripped again around 1540 (about a shift was lost).

· (2/20) Arc2A current was not stable. The regulator board was replaced (1.67 hrs.).
· (2/20) MAB6T01 mismatched. It was reset OK. It affected pass 3 beam (Hall A beam). The mismatch was not identified until 7 hours later (0.8 hrs.). Do not silent an alarm unless you are sure what it is and what action you should take.
· (2/17) Lambertson viewer did not show the video and the camera was replaced. (1.4 hrs.)

· (2/17) VBV1L13A would not open and the position indicator was replaced (1.6 hrs.)

· (2/16) iocea3 controller and the FSD card for 3E01 node were replaced (2 hrs.)
Hall A---Pass 1, E08 - 010
· Hall A beam is fine.

· (2/26) Hall A lost its target chamber vacuum during movement of the BigBite spectrometer. The RadCon rope used to define a rad area was attached to the spectrometer and the vacuum line. During the movement the vacuum line was damaged. Remember when   
· (2/25) Friday we magnetically separated 1-pass beam into Hall A since both B and C were down. The hall has been taking low current (<50uA) and the hall has been happy with beam. Hall A beam size is kind of large (more than 300x300 micrometers) but for what they are doing the size is an issue. 
· Beam is good. The hall will soon need high current (> 100uA).

· (2/17) Changed to pass 3. We forgot to initialize the fast raster and calibrate the ion chambers (they were taken care of three shifts later).
Hall B ---Pass 5, TPE
· The hall has been down since Friday preparing a new experiment (HDice).
· Beam is good. The hall will finish its current run Friday and HDice will start on March 26.

· (2/18) Changed to pass 5.

· (2/17) Changed to pass 2.
Hall C--- Pass 1, QWeak 
· Hall C has been down since Friday due to the Qtor power supply. The power supply was damaged when power was interrupted Friday. 

· Beam is good.

· (2/22)The width of Hall C’s beam charge asymmetry was larger than the specification. The optimization of the injector improved the beam quality.

· The hall started after the vacuum fix but the vacuum is not stable. They may ask beam to be off periodically to recover the vacuum.

· (2/18) Hall C was down for more than 5 shifts due to its vacuum issues at the end of its beam line (right upstream of its beam dump). 
